Author's response to reviews

Title: Clinical pharmacy activities in chronic kidney disease and end-stage renal disease patients: A systematic literature review

Authors:

Gunar Stemer (gunar.stemer@akhwien.at)
Rosa Lemmens-Gruber (rosa.lemmens@univie.ac.at)

Version: 5 Date: 20 July 2011

Author's response to reviews: see over
Cover Letter

Submission for publication:

Clinical pharmacy activities in chronic kidney disease and end-stage renal disease patients: A systematic literature review.

By Gunar Stemer and Rosa Lemmens-Gruber

Vienna, 20.07.2011

Dear Editor.

The following changes have been made:

1.) We apologize for the misunderstanding regarding the order of the references. Tables 2 and 3 were thought to be inserted in the manuscript and the references were organized in this way. Now they appear in numerical order.
2.) The repeated title was removed.
3.) All lines in the tables were removed.

Thank you very much for the feedback.

Sincerely,
Gunar Stemer